Author's response to reviews

Title: A randomized trial of mail vs. telephone invitation to a community-based cardiovascular health awareness program for older family practice patients [ISRCTN61739603]

Authors:

Tina Karwalajtys (karwalt@mcmaster.ca)
Janusz Kaczorowski (kaczorow@mcmaster.ca)
Larry W Chambers (lchamber@scohs.on.ca)
Cheryl Levitt (clevitt@mcmaster.ca)
Lisa Dolovich (idolovic@mcmaster.ca)
Bea McDonough (bmcdonou@city.hamilton.on.ca)
Christopher Patterson (pattec@hhsc.ca)
James E Williams (williamj@mcmaster.ca)

Version: 5  Date: 9 August 2005

Author's response to reviews:

BMC Family Practice Editors

T. Karwalajtys
Assistant Professor (PT) / Research Coordinator
Department of Family Medicine, McMaster University
1200 Main Street West, HSC-2V10
Hamilton ON L8N 3Z5

August 4, 2005

Dear Editors:

As per the email exchange copied below, I have revised items 3, 7, and 15 in the reference list.

Sincerely,

T. Karwalajtys, MA
On behalf of the CHAP Working Group

Enclosures

MS: 1307905281637738

A randomized trial of mail vs. telephone invitation to a community-based cardiovascular health awareness program for older family practice patients [ISRCTN61739603]

Tina Karwalajtys, Janusz Kaczorowski, Larry W Chambers, Cheryl Levitt, Lisa Dolovich, Bea McDonough, Christopher Patterson and James E Williams

BMC Family Practice

Dear Dr Karwalajtys,
I have just checked your query with Editorial Production and they say the following:

If it's just a normal issue then they don't need to worry about it, just the volume and page numbers will be fine e.g.


But if it's a supplement then they would need to format it like this:


I hope that helps.

With best wishes,

Alex

Alex J Pemberton PhD
Assistant Editor
BMC-Series Journals
BioMed Central Ltd.
Middlesex House
34-42 Cleveland St
London, W1T 4LB
Tel: +44 (0)20 7631 9103
Fax: +44 (0)20 7580 1938
e-mail: editorial@biomedcentral.com
Web: <http://www.biomedcentral.com/>
hello,

One more question -- I'm not clear on how to handle issue numbers in the reference format (refs 3, 7, 15).
Please advise.

Thank you,

Tina